Author's response to reviews

Title: Effects of Carissa opaca fruits extracts on oxidative pulmonary damages and fibrosis in rats

Authors:
sumaira sahreen (sumairasahreen@yahoo.com)
M R Khan (mrkhangau@yahoo.com)
Rahmat Khan (rahmatqu_81@yahoo.com)

Version: 5 Date: 21 November 2013

Author's response to reviews: see over
Dear Editor

BMC Complementary and Alternative Medicine

Dear Sir,

Subject: Refer to Effects of Carissa opaca fruit extracts on oxidative pulmonary damages and fibrosis in rats

Dear Sir,

Reference to above manuscript; enclosed please find a revised version for resubmission. English language was corrected with the help of senior professor and our colleges in USA. The changes were edited with red color in the manuscript. We would like to thanks for your careful reading of the manuscript. We hope that it will now be considered suitable for publication in your esteemed Journal.

Thank you for your attention.

Dr. Rahmat Ali Khan
Assistant Professor
University of Science and Technology Bannu
Email: rahmatgul_81@yahoo.com
Cell: +00923145796104